\begin{table}
\caption{\label{tab_results}Table of results. In Col.~(1) star identifier; Col.~(2) the spectral type (from SIMBAD), the other columns are described in Sect.~\ref{res}. The $EW$ of H$\alpha$ (Col.~5) is negative when in emission, ``a'' for absorption, and ``fi'' for filled up H$\alpha$. The evolutionary state of the circumstellar disk (Col.~14) is represented by ``accr.'' for accretion disk, ``non-accr.'' for non-accreting and optically thick disks, and ``debris'' for disks with cold dust only, ``none'' for verified absence of a circumstellar disk. No entry means no data is available. Stars marked with $^\dagger$ have been taken from the Torres et~al. (\cite{Torres2006}) list.} 
\begin{tabular}{llrrrrrcrrrrrll}
\hline\hline
(1) & (2) & (3) & (4) & (5) & (6) & (7) & (8) & (9) & (10) & (11) & (12) & (13) & (14) & (15) \\
Star ID   & SpT$^{v,w}$ & $v\sin i$ & $\sigma_{v\sin i}$ & $EW$(H$\alpha$) & $\sigma_{EW({\rm H}\alpha)}$ & $W_{{\rm H}\alpha}$ & $EW$(\ion{He}{i}) & Veiling & $EW$(\ion{Li}{i}) & $\sigma_{EW{\rm (\ion{Li}{i})}}$ & Age & $\sigma_{{\rm Age}}$ & disk\tablefootmark{b} & YSA\tablefootmark{v} \\
%\hline
& & km~s$^{-1}$ & km~s$^{-1}$ & \AA & \AA & \AA & \AA & & m\AA & m\AA & Myr & Myr & & \\
%\hline
\hline
\object{HD 105}  & G1 & 14.7 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 145 & 4 & 50 & 15 & debris & Tuc/Hor\tablefootmark{u} \\
\object{HD 225213}  & M1.5 & 5.8 & 1.7 & a & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & \ldots & field \\
\object{HD 377}  & G6 & 14.2 & 1.0 &  & \ldots & \ldots & \ldots & \ldots & 131 & 10 & 90 & 15 & debris & field\tablefootmark{b} \\
\object{HD 987}  & G5 & 7.3 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 191 & 5 & 30 & 15 & none\tablefootmark{p} & Tuc/Hor\tablefootmark{u}\\
\object{HD 984}  & F6 & 41.8 & 5.1 & a & \ldots & \ldots & \ldots & \ldots & 131 & 5 & 25 & 10 & none & field\tablefootmark{b} \\
\object{HD 1466}  & F8 & 22.7 & 2.2 & a & \ldots & \ldots & \ldots & \ldots & 117 & 3 & 30 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{HD 3221}  & K4.5 & 146.6 & 11.5 & --2.03 & 0.05 & 10.2 & \ldots & \ldots & 174 & 5 & 30 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{CD-78 24}$^\dagger$ & K3 & 30.7 & 2.3 & --0.1 & 0.01 & 1.5 & \ldots & \ldots & 291 & 8 & 15 & 10 & \ldots & field \\
\object{HD 4944}  & F8.5 & 33.5 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 132 & 5 & 30 & 10 & \ldots & field\tablefootmark{t} \\
\object{TYC 8852-264-1 } & G9 & 31.7 & 1.8 & --0.21 & 0.03 & 2.4 & \ldots & \ldots & 287 & 5 & 20 & 10 & none & Tuc/Hor\tablefootmark{u} \\
\object{HD 7661}  & G7.5 & 3.8 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 34 & 5 & 300 & 50 & none & field\tablefootmark{b} \\
\object{HIP 6276}  & G6 & 4.2 & 1.7 & a & \ldots & \ldots & \ldots & \ldots & 141 & 5 & 50 & 10 & debris & AB~Dor\tablefootmark{u} \\
\object{HD 8813}  & G3.5 & 6.7 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 120 & 4 & 120 & 20 & \ldots & field\tablefootmark{t} \\
\object{HD 8558}  & G5.5 & 13.6 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 189 & 3 & 30 & 15 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{CD-31 571}$^\dagger$ & K2 & 22.3 & 1.5 & fi & \ldots & \ldots & \ldots & \ldots & 260 & 5 & 30 & 10 & \ldots & field \\
\object{HD 9054}  & K1 & 3.9 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 182 & 7 & 30 & 10 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{TYC 8047-232-1}  & K2.5 & 20.6 & 1.2 & --0.11 & 0.01 & 1.8 & \ldots & \ldots & 333 & 5 & 30 & 10 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 12039}  & G2.5 & 15.8 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 173 & 4 & 30 & 15 & debris & Tuc/Hor\tablefootmark{u} \\
\object{1RXS J020154.6-523453}  & K3.5 & 19.9 & 1.2 & --0.18 & 0.02 & 2.0 & \ldots & \ldots & 298 & 5 & 20 & 10 & \ldots & Tuc/Hor\tablefootmark{u}\\
\object{HD 13246}  & F7.5 & 43.3 & 2.7 & a & \ldots & \ldots & \ldots & \ldots & 125 & 5 & 30 & 15 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 14706}  & G0 & 7.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 99 & 5 & 90 & 10 & \ldots & field \\
\object{HD 15526}  & G4 & 8.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 155 & 3 & 60 & 15 & none & field\tablefootmark{b} \\
\object{GSC 08056-00482}  & M2.5 & 29.3 & 1.7 & --4.98 & 0.6 & 3.5 & \ldots & \ldots & 350 & 5 & 20 & 10 & none\tablefootmark{p} & Tuc/Hor\tablefootmark{u} \\
\object{V* AF Hor} & M2.5 & 19.0 & 1.3 & --4.07 & 0.2 & 2.1 & \ldots & \ldots & 16 & 5 & 20 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{HIP 12545}  & K9.5 & 9.0 & 1.0 & --1.07 & 0.05 & 3.0 & \ldots & \ldots & 427 & 5 & 10 & 3 & none\tablefootmark{p} & $\beta$~Pic\tablefootmark{u} \\
\object{TYC 8497-995-1}  & K9 & 5.9 & 1.4 & --0.64 & 0.05 & 2.0 & \ldots & \ldots & 98 & 5 & 30 & 10 & \ldots & Tuc/Hor\tablefootmark{u}                   \\
\object{HD 17662}$^\dagger$ & G5 & 17.0 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 42 & 2 & 250 & 50 & \ldots & field \\
\object{HD 17925}  & K0 & 4.8 & 1.6 & a & \ldots & \ldots & \ldots & \ldots & 206 & 5 & 90 & 10 & \ldots & field\tablefootmark{b} \\
\object{CD-37 1123}$^\dagger$ & G9 & 6.9 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 230 & 5 & 30 & 15 & \ldots & field \\
\object{HD 19330}  & F9 & 7.0 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 18 & 5 & 400 & 50 & \ldots & \ldots \\
\object{HD 19491} & G1 & 5.2 & 1.5 & a & \ldots & \ldots & \ldots & \ldots & 33 & 5 & 600 & 50 & \ldots & \ldots \\
\object{HD 19668}  & G5.5 & 7.1 & 1.1 &  & \ldots & \ldots & \ldots & \ldots & 165 & 5 & 90 & 10 & debris & field\tablefootmark{b} \\
\object{HD 21411}  & G6 & 4.1 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & $<$2 & \ldots & $>$600 & \ldots & none & field\tablefootmark{b} \\
\object{TYC 5879-490-1}  & G4 & 14.7 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 188 & 4 & 30 & 15 & none & field\tablefootmark{b} \\
\object{RX J 0329.1+0118}  & F9 & 65.1 & 4.0 & a & \ldots & \ldots & \ldots & \ldots & 82 & 5 & 200 & 50 & none & field\tablefootmark{b} \\
\object{1RXS J033149.8-633155}$^\dagger$ & G9 & 18.1 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 300 & 3 & 25 & 15 & \ldots & field \\
\object{HD 21955}$^\dagger$ & G7 & 27.7 & 1.7 & a & \ldots & \ldots & \ldots & \ldots & 230 & 1 & 30 & 10 & \ldots & field \\
\object{HD 23208}$^\dagger$ & G8 & 9.3 & 0.4 & a & \ldots & \ldots & \ldots & \ldots & 252 & 3 & 25 & 10 & \ldots & \ldots \\
\object{RX J 0354.4+0535}  & G2 & 69.6 & 5.6 & a & \ldots & \ldots & \ldots & \ldots & 118 & 5 & 90 & 10 & none & field\tablefootmark{b} \\
\object{HD 25300}  & K4 & 13.0 & 1.0 & --0.62 & 0.05 & 4.0 & \ldots & \ldots & 108 & 5 & 90 & 10 & none & field\tablefootmark{b}\\
\object{TYC 5882-1169-1}  & K2 & 6.5 & 1.3 & a & \ldots & \ldots & \ldots & \ldots & 221 & 5 & 30 & 15 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 25457}$^\dagger$ & F6.5 & 18.5 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 112 & 3 & 50 & 15 & \ldots & AB~Dor\tablefootmark{u} \\
\object{CD-31 1688}$^\dagger$ & G6 & 30.0 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & 240 & 4 & 30 & 10 & \ldots & field \\
\object{HD 26990}  & G2 & 7.1 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 47 & $<$1 & 300 & 50 & none & field\tablefootmark{r} \\
\object{CD-43 1395}$^\dagger$ & G7 & 28.1 & 1.7 & a & \ldots & \ldots & \ldots & \ldots & 270 & 4 & 30 & 10 & \ldots & field \\
\object{CD-43 1451}$^\dagger$ & G9 & 21.3 & 1.4 & fi & \ldots & \ldots & \ldots & \ldots & 280 & 4 & 15 & 5 & \ldots & field \\
\object{TYC 5891-69-1}$^\dagger$ & G4 & 20.5 & 1.3 & fi & \ldots & \ldots & \ldots & \ldots & 280 & 4 & 10 & 5 & none & field \\
\object{1RXS J043451.0-354715}$^\dagger$ & K1 & 11.3 & 1.0 & --0.24 & 0.03 & 1.9 & \ldots & \ldots & 300 & 5 & 20 & 15 & \ldots & field \\
\object{RX J 0434.3+0226}  & K5.5 & 5.7 & 1.4 & --0.67 & 0.03 & 2.2 & \ldots & \ldots & 233 & 3 & 30 & 10 & none & field\tablefootmark{b} \\
\object{HD 29623}  & G2 & 8.8 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 96 & 1 & 90 & 10 & \ldots & field\tablefootmark{t} \\
\object{RX J 0442.5+0906}  & K0 & 12.1 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 240 & 5 & 60 & 15 & none & field\tablefootmark{b} \\
\object{HIP 23309}  & K7 & 5.8 & 1.5 & --1.21 & 0.03 & 2.3 & \ldots & \ldots & 341 & 3 & 10 & 3 & none\tablefootmark{p} & $\beta$~Pic\tablefootmark{u} \\
\object{HD 31950}$^\dagger$ & G & 29.2 & 0.8 & a & \ldots & \ldots & \ldots & \ldots & 181 & 6 & 30 & 15 & none\tablefootmark{b} & field \\
\object{GJ 3323}  & M1.5 & 10.1 & 3.8 & --5.43 & 0.2 & 2.7 & \ldots & \ldots & 11 & 5 & 12 & 15 & \ldots & $\beta$~Pic\tablefootmark{u} \\
\object{GJ 3323B}  & \ldots & 11.3 & 1.5 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HD 35114}  & F7 & 73.8 & 5.7 & a & \ldots & \ldots & \ldots & \ldots & 108 & 1 & 30 & 10 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 35850}  & G0 & 71.6 & 9.6 & a & \ldots & \ldots & \ldots & \ldots & 149 & 10 & 12 & 15 & \ldots & $\beta$~Pic\tablefootmark{u} \\
\object{V* AB Dor} & K1 & 104.5 & 10.9 & a & \ldots & \ldots & \ldots & \ldots & 233 & 19 & 50 & 15 & none\tablefootmark{c} & AB~Dor\tablefootmark{u} \\
\object{V* GW Ori}$^\dagger$ & K3 & 43.7 & 2.4 & --20.76 & 0.2 & 9.9 & 0.05 & 0.95 & 240 & 5 & 5 & 2 & accr.\tablefootmark{e} & Ori \\
\object{TYC 7600-0516-1}  & K0.5 & 18.6 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 248 & 5 & 30 & 15 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 37484}  & F3 & 48.4 & 4.0 & a & \ldots & \ldots & \ldots & \ldots & 92 & 3 & 30 & 15 & \ldots & field\tablefootmark{b} \\
\object{V* AI Lep} & G0 & 27.9 & 1.6 & a & \ldots & \ldots & \ldots & \ldots & 200 & 7 & 30 & 15 & \ldots & field\tablefootmark{b} \\
\object{HD 38397}  & G1 & 15.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 151 & 4 & 60 & 15 & \ldots & field\tablefootmark{t} \\
\object{HD 38207}  & F2 & 70.7 & 4.2 & a & \ldots & \ldots & \ldots & \ldots & 75 & 5 & 55 & 15 & \ldots & field\tablefootmark{b} \\
\object{HD 38949}  & G1 & 7.2 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 82 & $<$1 & 250 & 50 & \ldots & field\tablefootmark{b} \\
\object{HD 42270}$^\dagger$ & K0 & 32.6 & 2.3 & a & \ldots & \ldots & \ldots & \ldots & 305 & 1 & 20 & 10 & \ldots & field \\
\object{HD 40216}  & F7 & 32.2 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & 115 & 2 & 30 & 10 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{HD 41700}  & F8.5 & 16.6 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 78 & 5 & 200 & 50 & \ldots & field\tablefootmark{r} \\
\object{CD-34 2676}$^\dagger$ & K1 & 18.1 & 1.1 & \ldots & \ldots & \ldots & \ldots & \ldots & 240 & 5 & 65 & 15 & \ldots & field \\
\object{HD 45081}  & K3.5 & 15.9 & 1.0 & --0.79 & 0.02 & 2.1 & \ldots & \ldots & 398 & 3 & 10 & 5 & none & $\beta$~Pic\tablefootmark{u} \\
\object{V* AB Pic} & K0.5 & 11.5 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 287 & 4 & 30 & 15 & none & Tuc/Hor\tablefootmark{u} \\
\object{HD 43989}  & G0.5 & 42.3 & 5.0 & \ldots & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & debris & Tuc/Hor\tablefootmark{u} \\
\object{HD 45270}  & G0 & 17.2 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 139 & 5 & 50 & 10 & \ldots & AB~Dor\tablefootmark{u} \\
\object{HD 47875}  & G3 & 11.6 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 198 & 3 & 30 & 15 & none & field\tablefootmark{b} \\
\object{HD 48189}  & G0 & 16.5 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 132 & 3 & 50 & 10 & none\tablefootmark{c} & AB~Dor\tablefootmark{u} \\
\object{HD 51062}$^\dagger$ & G5 & 14.2 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 180 & 2 & 200 & 50 & \ldots & Carina-Vela \\
\object{TYC 8545-1758-1}  & F9 & 46.7 & 6.8 & a & \ldots & \ldots & \ldots & \ldots & 136 & 5 & 25 & 10 & \ldots & field\tablefootmark{b} \\
\object{HD 51797}$^\dagger$ & K0 & 14.5 & 0.5 & a & \ldots & \ldots & \ldots & \ldots & 288 & 5 & 30 & 15 & \ldots & \ldots \\
\object{HD 55279}  & K2 & 9.4 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 271 & 4 & 30 & 15 & \ldots & Tuc/Hor\tablefootmark{u} \\
\object{CD-84 80}$^\dagger$ & G9 & 7.7 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 300 & 5 & 30 & 15 & \ldots & field \\
\object{HD 61005}  & G5.5 & 9.9 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 169 & 3 & 30 & 15 & debris & field\tablefootmark{r} \\
\object{HD 62850}  & F9.5 & 14.4 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 122 & 1 & 90 & 10 & \ldots & field\tablefootmark{r}\\
\object{HD 70573}  & G2 & 13.5 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 148 & 4 & 70 & 15 & \ldots & field\tablefootmark{b} \\
\object{V* EG Cha}$^\dagger$ & K4 & 21.4 & 1.3 & --0.96 & 0.05 & 2.9 & \ldots & \ldots & 510 & 6 & 5 & 2 & none\tablefootmark{d} & Cha \\
\object{RX J 0849.2-7735}  & K2 & 6.0 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & 120 & 4 & 90 & 10 & none & Cha\tablefootmark{f} \\
\object{HD 75393}  & F9 & 4.7 & 1.6 & a & \ldots & \ldots & \ldots & \ldots & 9 & 2 & $>$600 & 50 & none & field\tablefootmark{b} \\
\object{RX J 0850.1-7554}  & G7 & 47.7 & 3.6 & a & \ldots & \ldots & \ldots & \ldots & 210 & 5 & 60 & 15 & none & Cha\tablefootmark{f} \\
\object{RX J 0917.2-7744}  & G4.5 & 76.5 & 8.9 & a & \ldots & \ldots & \ldots & \ldots & 165 & 5 & 30 & 10 & none & Cha\tablefootmark{f}\\
\object{HD 81544}$^\dagger$ & G9 & 12.9 & 0.5 & fi & \ldots & \ldots & \ldots & \ldots & 331 & 5 & 8 & 3 & \ldots & \ldots \\
\object{HD 84323}  & F9 & \ldots & \ldots & a & \ldots & \ldots & \ldots & \ldots & 197 & 5 & 30 & 15 & \ldots & field \\
\object{TYC 7697-2254-1}$^\dagger$ & G8 & 9.0 & 0.4 & fi & \ldots & \ldots & \ldots & \ldots & 240 & 4 & 25 & 10 & \ldots & \ldots \\
\object{HD 86356}  & G8 & 136.4 & 10.8 & --0.04 & 0.01 & 1.5 & \ldots & \ldots & 265 & 5 & 10 & 5 & none & Cha\tablefootmark{f} \\
\object{HD 86021}$^\dagger$ & G7 & 24.3 & 1.6 & a & \ldots & \ldots & \ldots & \ldots & 78 & 10 & 200 & 100 & \ldots & \ldots \\
\object{TYC 6604-118-1}  & K1 & 18.3 & 1.1 & --0.11 & 0.01 & 1.5 & \ldots & \ldots & 72 & 5 & 8 & 5 & \ldots & TWA\tablefootmark{k} \\
\object{TYC 6604-118-1B}  & \ldots & 65.4 & 6.0 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HD 88201}  & G0 & 10.3 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 71 & 4 & 300 & 50 & none & field\tablefootmark{b} \\
\object{TWA 21}  & K2 & 8.2 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 357 & 3 & 8 & 3 & \ldots & TWA\tablefootmark{u} \\
\object{TWA 6}  & K6 & 94.7 & 9.1 & --3.55 & 0.6 & 5.1 & \ldots & \ldots & 500 & 5 & 7 & 3 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{HD 90712}  & G1.5 & 11.2 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 99 & 1 & 250 & 50 & none & field\tablefootmark{b}\\
\object{RX J 1029.5-6349}  & K1.5 & 24.3 & 1.5 & a & \ldots & \ldots & \ldots & \ldots & 251 & 5 & 40 & 15 & none & IC~2602\tablefootmark{b} \\
\object{RX J 1040.0-6315}  & G2.5 & 13.9 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 177 & 6 & 30 & 15 & none & IC~2602\tablefootmark{b} \\
\object{HD 307938}  & F9 & 81.1 & 5.0 & a & \ldots & \ldots & \ldots & \ldots & 161 & 1 & 30 & 15 & \ldots & IC~2602\tablefootmark{b} \\
\object{TWA 7}  & M1 & 5.0 & 2.1 & --4.64 & 0.5 & 2.1 & \ldots & \ldots & 465 & 10 & 8 & 3 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{RX J 1046.2-6402}  & G3 & 35.2 & 2.0 & a & \ldots & \ldots & \ldots & \ldots & 187 & 5 & 40 & 10 & none & IC~2602\tablefootmark{b} \\
\object{2MASS J10574936-6913599}  & K3 & 26.8 & 4.2 & a & \ldots & \ldots & \ldots & \ldots & 404 & 8 & 5 & 3 & none & ScoCen/LCC\tablefootmark{b} \\
\object{V* CR Cha}$^\dagger$ & K2 & 36.5 & 4.4 & --26.94 & 1.24 & 10.55 & 0.06 & 0.07 & 380 & 3 & 5 & 3 & accr. & Cha \\
\object{V* TW Hya} & M0 & 6.2 & 1.3 & --116.78 & 5 & 9.5 & 1.94 & 0.8 & 361 & $<$1 & 8 & 3 & accr.\tablefootmark{e} & TWA\tablefootmark{u} \\
\object{HD 96064}  & G7 & 6.1 & 1.3 & a & \ldots & \ldots & \ldots & \ldots & 106 & 2 & 90 & 10 & \ldots & field\tablefootmark{t} \\
\object{V* DI Cha}$^\dagger$ & G2 & 38.1 & 2.2 & --26.07 & 0.5 & 9.8 & 0.03 & 0.05 & 260 & 1 & 5 & 2 & accr.\tablefootmark{g} & Cha \\
\object{TWA 2}  & M0.5 & 11.8 & 1.0 & --1.89 & 0.07 & 2.2 & \ldots & \ldots & 509 & 11 & 8 & 3 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{HD 98553}  & G2 & 4.0 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 21 & 1 & 300 & 50 & none & field\tablefootmark{b} \\
\object{TWA 4}  & K1 & 5.0 & 1.6 & --0.09 & 0.01 & 1.5 & \ldots & \ldots & 312 & 5 & 8 & 5 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{TWA 4B}  & \ldots & 8.0 & 1.1 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HD 99409}  & G6 & 17.6 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 95 & $<$1 & 125 & 20 & \ldots & field\tablefootmark{t} \\
\object{TWA 5}  & M1.5 & 63.2 & 9.5 & --8.5 & 0.3 & 8.8 & \ldots & \ldots & 564 & 5 & 8 & 3 & none\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{TWA 8}  & M1 & 7.1 & 1.2 & --6.43 & 0.5 & 2.8 & \ldots & \ldots & 475 & 5 & 8 & 3 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{RX J 1140.3-8321}  & K1 & 9.3 & 1.1 & --0.18 & 0.02 & 2.1 & \ldots & \ldots & 195 & 1 & 15 & 10 & none & Cha\tablefootmark{f} \\
\object{HD 101472}  & F8.5 & 5.8 & 1.3 & a & \ldots & \ldots & \ldots & \ldots & 112 & 2 & 90 & 10 & none & field\tablefootmark{b} \\
\object{TWA 19}  & G7 & 30.1 & 1.7 & a & \ldots & \ldots & \ldots & \ldots & 197 & 5 & 8 & 5 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{V* GQ Leo} & K8 & 7.2 & 1.6 & --0.63 & 0.02 & 2.2 & \ldots & \ldots & 83 & 2 & 12 & 4 & none & field \\
\object{TWA 9}  & K4 & 10.4 & 1.0 & --1.57 & 0.07 & 2.1 & \ldots & \ldots & 506 & 5 & 7 & 3 & non-accr.\tablefootmark{i} & TWA\tablefootmark{u} \\
\object{V* T Cha}$^\dagger$ & G4 & 40.0 & 2.7 & fi & \ldots & \ldots & \ldots & \ldots & 382 & 1 & 4 & 2 & non-accr.\tablefootmark{a} & Cha \\
\object{HD 104576}  & G1 & 3.5 & 2.1 & a & \ldots & \ldots & \ldots & \ldots & 52 & 2 & 300 & 50 & none & field\tablefootmark{b}\\
\object{RX J 1203.7-8129}  & K2 & 2.7 & 2.7 & a & \ldots & \ldots & \ldots & \ldots & 69 & 6 & 250 & 50 & none & Cha\tablefootmark{f}? \\
\object{TWA 24}  & K1.5 & 15.5 & 1.0 & --0.6 & 0.02 & 2.8 & \ldots & \ldots & 383 & 5 & 7 & 3 & \ldots & TWA\tablefootmark{u} \\
\object{HD 105690}  & G4 & 8.5 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 149 & 1 & 8 & 15 & \ldots & TWA\tablefootmark{k} \\
\object{2MASS J12123577-5520273}  & K4 & 28.5 & 1.6 & --0.2 & 0.02 & 1.5 & \ldots & \ldots & 358 & 2 & 5 & 3 & debris & ScoCen/LCC\tablefootmark{b} \\
\object{2MASS J12143410-5110124}  & K0 & 17.9 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 311 & 5 & 25 & 10 & \ldots & ScoCen/LCC\tablefootmark{b} \\
\object{TWA 25}  & M0 & 11.6 & 1.1 & --1.87 & 0.6 & 2.8 & \ldots & \ldots & 531 & 13 & 8 & 3 & \ldots & TWA\tablefootmark{u} \\
\object{RX J 1220.6-7539}  & K2.5 & 6.3 & 1.3 & --0.2 & 0.03 & 2.1 & \ldots & \ldots & 208 & 2 & 15 & 10 & none & Cha\tablefootmark{f} \\
\object{HD 107441}  & K2 & 41.5 & 2.7 & a & \ldots & \ldots & \ldots & \ldots & 226 & 4 & 30 & 10 & none & ScoCen/LCC\tablefootmark{b} \\
\object{2MASS J12223322-5333489}  & G1 & 70.9 & 7.3 & a & \ldots & \ldots & \ldots & \ldots & 191 & 5 & 30 & 10 & debris & ScoCen/LCC\tablefootmark{b} \\
\object{RX J 1225.3-7857}   & G7 & 8.2 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 119 & 5 & 125 & 20 & none & Cha\tablefootmark{f} \\
\object{HD 108799}  & G0 & 7.2 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 94 & 2 & 125 & 20 & none & field\tablefootmark{b}\\
\object{HD 109138}$^\dagger$ & K0 & 6.4 & 0.4 & a & \ldots & \ldots & \ldots & \ldots & 129 & 3 & 90 & 15 & \ldots & \ldots \\
\object{1RXS J123332.4-571345}$^\dagger$ & K1 & 18.2 & 0.8 & fi & \ldots & \ldots & \ldots & \ldots & 424 & 8 & 8 & 5 & \ldots & \ldots \\
\object{TYC 9412-59-1}$^\dagger$ & K3 & 18.0 & 0.8 & fi & \ldots & \ldots & \ldots & \ldots & 429 & 8 & 5 & 3 & \ldots & Cha \\
\object{TYC 8654-1115-1}$^\dagger$ & G9 & 19.4 & 0.4 & a & \ldots & \ldots & \ldots & \ldots & 298 & 7 & 5 & 3 & \ldots & \ldots \\
\object{HD 111170}  & G9 & 133.2 & 7.7 & a & \ldots & \ldots & \ldots & \ldots & 4 & 1 & $>$600 & 50 & none & ScoCen/LCC\tablefootmark{b} \\
\object{TYC 8655-149-1}  & G7.5 & 28.0 & 1.6 & a & \ldots & \ldots & \ldots & \ldots & 235 & 5 & 30 & 10 & none & ScoCen/LCC\tablefootmark{b} \\
\object{CD-69 1055}$^\dagger$ & K0 & 26.0 & 1.5 & fi & \ldots & \ldots & \ldots & \ldots & 400 & 6 & 5 & 3 & \ldots & LCC \\
\object{2MASS J13015069-5304581}  & K1.5 & 8.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 367 & 5 & 8 & 3 & debris & ScoCen/LCC\tablefootmark{b} \\
\object{HD 113553}  & G4 & 11.6 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 139 & 1 & 90 & 10 & \ldots & field\tablefootmark{r} \\
\object{TYC 8649-251-1}  & F9.5 & 115.8 & 10.2 & a & \ldots & \ldots & \ldots & \ldots & 208 & 5 & 30 & 15 & none & ScoCen/LCC\tablefootmark{b} \\
\object{V* MP Mus} & K2 & 12.9 & 1.0 & --42.11 & 3 & 10.5 & 0.5 & 0.1 & 410 & 5 & 7 & 3 & accr. & ScoCen/LCC\tablefootmark{b} \\
\object{HD 117524}  & G7 & 32.1 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 210 & 3 & 30 & 10 & none & ScoCen/LCC\tablefootmark{b} \\
\object{2MASS J13375730-4134419}  & K1 & 13.7 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 312 & 3 & 30 & 10 & debris & ScoCen/UCL\tablefootmark{b} \\
\object{HD 119269}  & G5 & 24.2 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & 211 & 4 & 30 & 10 & \ldots & ScoCen/LCC\tablefootmark{b} \\
\object{TYC 8270-2015-1}  & K0 & 69.6 & 8.7 & --0.17 & 0.02 & 1.8 & \ldots & \ldots & 232 & 5 & 30 & 10 & \ldots & ScoCen/UCL\tablefootmark{b} \\
\object{HD 120812}  & F9.5 & 25.9 & 1.5 & a & \ldots & \ldots & \ldots & \ldots & 94 & 5 & 30 & 15 & none & ScoCen/UCL\tablefootmark{b} \\
\object{TYC 7811-2909-1}  & K1 & 17.5 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 329 & 5 & 25 & 10 & none & ScoCen\tablefootmark{b} \\
\object{HD 124784}$^\dagger$ & G0 & 17.3 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 140 & 1 & 200 & 50 & \ldots & UCL \\
\object{HD 124784B}  & \ldots & 17.5 & 1.1 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{TYC 8282-516-1}  & G9 & 10.3 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 290 & 5 & 30 & 10 & debris & ScoCen/UCL\tablefootmark{b} \\
\object{HD 126670}  & G7 & 13.4 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 275 & 2 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 128242}  & G1 & 115.8 & 6.7 & a & \ldots & \ldots & \ldots & \ldots & 60 & 12 & 300 & 50 & none & ScoCen/UCL\tablefootmark{b} \\
\object{RX J 1457.3-3613}  & G5 & 31.9 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & 280 & 3 & 20 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 132173}  & F9 & 9.6 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 117 & 2 & 30 & 10 & \ldots & field\tablefootmark{b} \\
\object{TYC 8297-1613-1}  & G4 & 17.9 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 192 & 5 & 30 & 10 & \ldots & ScoCen/UCL\tablefootmark{b} \\
\object{RX J 1507.2-3505}   & K0.5 & 12.2 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 320 & 5 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 133938}  & G7.5 & 24.9 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & 223 & 4 & 45 & 15 & none & ScoCen/UCL\tablefootmark{b} \\
\object{V* LT Lup}$^\dagger$ & K0 & 27.8 & 1.7 & fi & \ldots & \ldots & \ldots & \ldots & 385 & 5 & 8 & 3 & none\tablefootmark{o} & Lup \\
\object{RX J 1518.4-3738}   & K2.5 & 25.2 & 1.5 & a & \ldots & \ldots & \ldots & \ldots & 321 & 5 & 30 & 15 & none & ScoCen/UCL\tablefootmark{b} \\
\object{V* LY Lup}$^\dagger$ & K0 & 17.2 & 1.1 & --0.5 & 0.05 & 2.5 & \ldots & \ldots & 401 & 10 & 3 & 2 & none\tablefootmark{o} & Lup \\
\object{V* MP Lup}$^\dagger$ & K0 & 18.0 & 1.3 & fi & \ldots & \ldots & \ldots & \ldots & 360 & 2 & 5 & 3 & none\tablefootmark{o} & Lup \\
\object{V* MS Lup}$^\dagger$ & G7 & 28.9 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & 308 & 6 & 5 & 3 & none\tablefootmark{o} & Lup \\
\object{RX J 1531.3-3329}   & G9 & 15.6 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 281 & 5 & 30 & 15 & none & ScoCen/UCL\tablefootmark{l} \\
\object{1RXS J153328.4-665130}\footnote{{\scriptsize \object{TYC 9034-968-1}}}$^\dagger$ & K2 & 9.1 & 1.1 & fi & \ldots & \ldots & \ldots & \ldots & 295 & 1 & 30 & 15 & \ldots & field \\
\object{HD 139084}  & G8 & 18.2 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 278 & 4 & 50 & 15 & none & $\beta$~Pic\tablefootmark{u} \\
\object{HD 139498}  & G7 & 32.6 & 3.2 & a & \ldots & \ldots & \ldots & \ldots & 149 & 5 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 139498B}  & \ldots & 17.4 & 1.1 &  & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HD 140374}  & G7.5 & 5.1 & 1.5 & a & \ldots & \ldots & \ldots & \ldots & 124 & 5 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 140374B}  & \ldots & 8.1 & 1.1 &  & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{RX J 1544.0-3311}  & K1 & 10.3 & 1.0 & --1.25 & 0.03 & 3.7 & \ldots & \ldots & 357 & 3 & 12 & 3 & none & ScoCen/UCL\tablefootmark{b} \\
\object{V* HT Lup}$^\dagger$ & K2 & 44.8 & 3.0 & --7.02 & 0.2 & 11.0 & \ldots & \ldots & 442 & 5 & 5 & 2 & non-accr. & Lup \\
\object{V* GQ Lup}$^\dagger$ & K8 & 9.1 & 1.1 & --39.53 & 0.5 & 13.2 & 0.6 & 0.5 & 644 & 5 & 2 & 3 & accr. & Lup \\
\object{HD 141521} & G3.5 & 7.1 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 149 & 2 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{HD 141521B}  & \ldots & 8.1 & 1.1 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HBC 603}$^\dagger$ & M0 & 5.9 & 0.6 & --16.08 & 2.5 & 7.71 & 0.1 & 0.05 & 566 & 10 & 5 & 3 & accr. & \ldots \\
\object{HD 141943}  & G4 & 36.3 & 2.1 & a & \ldots & \ldots & \ldots & \ldots & 198 & 2 & 30 & 15 & debris & field\tablefootmark{b} \\
\object{HD 142229}  & G0 & 5.6 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & 116 & 5 & 90 & 10 & none & field\tablefootmark{b} \\
\object{HD 142361}  & G3 & 79.4 & 4.6 & a & \ldots & \ldots & \ldots & \ldots & 195 & 5 & 30 & 10 & none & ScoCen/UCL\tablefootmark{b} \\
\object{V* IM Lup}$^\dagger$ & M0 & 12.6 & 0.6 & --5.71 & 0.51 & 5.5 & 0.03 & 0.1 & 598 & 10 & 5 & 3 & accr. & Lup \\
\object{HD 143358}  & G1 & 72.5 & 4.4 & a & \ldots & \ldots & \ldots & \ldots & 166 & 3 & 30 & 10 & \ldots & ScoCen/UCL\tablefootmark{b} \\
\object{EM* SR 9}$^\dagger$ & K5 & 13.4 & 0.7 & --20.38 & 1.42 & 7.82 & 0.3 & 0.15 & 567 & 8 & 5 & 3 & accr. & \ldots \\
\object{V* V2129 Oph}$^\dagger$ & K3 & 13.5 & 1.0 & --14.6 & 2 & 6.5 & 0.25 & 0.15 & 530 & 5 & 3 & 2 & accr. & Oph \\
\object{DoAr 44}$^\dagger$ & K3 & 14.6 & 1.1 & --22.87 & 4.42 & 9.98 & 0.6 & 0.15 & 507 & 8 & 5 & 3 & accr. & \ldots \\
\object{V* V1121 Oph}$^\dagger$ & K5 & 8.4 & 0.5 & --20.81 & 2.48 & 8.28 & 0.3 & 0.25 & 550 & 6 & 8 & 3 & accr. & Oph \\
\object{HD 151798}  & G1 & 11.3 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 131 & 2 & 90 & 10 & none & field\tablefootmark{b} \\
\object{HD 152555}  & F9 & 17.8 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 123 & 2 & 50 & 10 & none & AB~Dor\tablefootmark{u} \\
\object{TYC 8726-57-1}$^\dagger$ & G9 & 22.1 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & 250 & 4 & 25 & 15 & \ldots & field \\
\object{CD-27 11535}$^\dagger$ & K5 & 12.3 & 1.1 & --1.06 & 0.07 & 3.1 & \ldots & \ldots & 490 & 1 & 6 & 3 & \ldots & field \\
\object{TYC 6812-348-1}$^\dagger$ & K3 & 19.1 & 1.4 & --0.27 & 0.05 & 2.0 & \ldots & \ldots & 455 & 4 & 5 & 2 & \ldots & field \\
\object{HD 155555}  & K2 & 39.9 & 2.3 & a & \ldots & \ldots & \ldots & \ldots & 170 & 5 & 12 & 3 & none\tablefootmark{p} & $\beta$~Pic\tablefootmark{u} \\
\object{HD 155555B}  & \ldots & 8.8 & 0.9 & \ldots & \ldots & \ldots & \ldots & \ldots & 169 & 5 & \ldots & \ldots & \ldots & \ldots \\
\object{HD 161460}$^\dagger$ & K0 & 10.1 & 1.0 & fi & \ldots & \ldots & \ldots & \ldots & 320 & 4 & 30 & 15 & \ldots & field \\
\object{RX J 1841.8-3525}  & G7 & 41.2 & 2.3 & a & \ldots & \ldots & \ldots & \ldots & 248 & 1 & 30 & 15 & none & Cor~Aus\tablefootmark{b}  \\
\object{HD 174656}  & G4 & 27.8 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 338 & 5 & 8 & 3 & none & Cor~Aus\tablefootmark{b}\\
\object{HD 181327}  & F5.5 & 19.4 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 108 & $<$1 & 15 & 10 & debris\tablefootmark{p} & $\beta$~Pic\tablefootmark{u}\\
\object{HD 183414}  & G3 & 9.6 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 142 & 3 & 75 & 15 & \ldots & field\tablefootmark{r} \\
\object{HD 191089}  & F9 & 48.8 & 3.0 & a & \ldots & \ldots & \ldots & \ldots & 101 & 5 & 125 & 20 & \ldots & field\tablefootmark{b} \\
\object{HD 193464}  & F9 & 15.0 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 92 & 2 & 300 & 50 & \ldots & field\tablefootmark{t} \\
\object{HD 197481}  & M0.5 & 8.2 & 1.2 & --3.16 & 0.05 & 2.8 & \ldots & \ldots & 56 & 3 & 15 & 10 & debris\tablefootmark{p} & $\beta$~Pic\tablefootmark{u} \\
\object{HD 199143}  & F9.5 & 161.5 & 11.8 & a & \ldots & \ldots & \ldots & \ldots & 60 & 5 & 300 & 50 & none & $\beta$~Pic\tablefootmark{u} \\
\object{HD 200746}  & G1.5 & 6.4 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 78 & 5 & 300 & 50 & none & field\tablefootmark{b} \\
\object{HD 201219}  & G2.5 & 3.5 & 2.1 & a & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & debris & field\tablefootmark{b} \\
\object{HD 201989}  & G2 & 5.0 & 1.5 & \ldots & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & none & field\tablefootmark{b} \\
\object{HD 202917}  & G4.5 & 14.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 211 & 1 & 30 & 15 & \ldots & Tuc/Hor\tablefootmark{u}\\
\object{HD 202947}  & K1 & 12.9 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 161 & 11 & 30 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u}\\
\object{HD 205905}  & K1 & 4.8 & 1.6 & \ldots & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & none & field\tablefootmark{b} \\
\object{HD 207278}$^\dagger$ & G7 & 11.2 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 190 & 1 & 50 & 10 & \ldots & AB~Dor \\
\object{HD 207575}  & F9 & 36.5 & 2.1 & a & \ldots & \ldots & \ldots & \ldots & 111 & 7 & 30 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{HD 209253}  & F8 & 16.9 & 1.0 & \ldots & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & \ldots & field\tablefootmark{b} \\
\object{HD 209779}  & G0 & 5.5 & 1.4 & a & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & none & field\tablefootmark{b} \\
\object{HD 212291}  & G4 & 1.5 & 4.4 & a & \ldots & \ldots & \ldots & \ldots & n/a & \ldots & \ldots & \ldots & none & field\tablefootmark{b} \\
\object{1RXS J223929.1-520525}$^\dagger$ & K0 & 9.4 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 245 & 1 & 60 & 15 & \ldots & field \\
\object{CP-72 2713}$^\dagger$ & K7 & 6.6 & 1.4 & --12.5 & 0.6 & 2.3 & \ldots & \ldots & 440 & 4 & 8 & 1 & \ldots & field \\
\object{CD-40 14901}$^\dagger$ & G5 & 14.9 & 1.0 & a & \ldots & \ldots & \ldots & \ldots & 175 & 2 & 75 & 15 & \ldots & field \\
\object{HD 216803}  & K3 & 4.5 & 1.8 & a & \ldots & \ldots & \ldots & \ldots & 26 & 5 & 400 & 50 & \ldots & field\tablefootmark{b} \\
\object{HD 217343}  & G3 & 12.2 & 0.9 & a & \ldots & \ldots & \ldots & \ldots & 163 & 1 & 45 & 10 & none & AB~Dor\tablefootmark{u} \\
\object{HD 217897}$^\dagger$ & M2.5 & 8.5 & 1.2 & a & \ldots & \ldots & \ldots & \ldots & 50 & 3 & 30 & 15 & \ldots & field \\
\object{BD-03 5579}  & K4 & 14.0 & 1.0 & --0.7 & 0.02 & 1.3 & \ldots & \ldots & 200 & 3 & 50 & 15 & \ldots & field \\
\object{TYC 9129-1361-1}  & K1 & 86.1 & 9.8 & a & \ldots & \ldots & \ldots & \ldots & $<$2 & 5 & $>$600 & 50 & \ldots & field \\
\object{TYC 9129-1361-1B}  & \ldots & 119.5 & 10.0 & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots & \ldots \\
\object{HD 220054}  & G8 & 31.8 & 1.9 & a & \ldots & \ldots & \ldots & \ldots & 259 & 4 & 30 & 10 & \ldots & field \\
\object{TYC 584-343-1}$^\dagger$ & K0 & 14.0 & 1.0 & fi & \ldots & \ldots & \ldots & \ldots & 300 & 2 & 30 & 15 & \ldots & field \\
\object{BD-13 6424}$^\dagger$ & M0 & 9.6 & 1.2 & --1.2 & 0.03 & 1.9 & \ldots & \ldots & 185 & 8 & 30 & 15 & \ldots & field \\
\object{HD 222259}  & G4 & 17.5 & 1.1 & a & \ldots & \ldots & \ldots & \ldots & 204 & 5 & 30 & 10 & debris\tablefootmark{m} & Tuc/Hor\tablefootmark{u} \\
\object{HD 224228}  & K2 & 2.7 & 2.8 & a & \ldots & \ldots & \ldots & \ldots & 60 & 5 & 50 & 20 & \ldots & AB~Dor\tablefootmark{u} \\
\hline
\end{tabular}
\tablebib{\tablefoottext{a}{Brown et~al. (\cite{Brown2007});} 
\tablefoottext{b}{Carpenter et~al. (\cite{Carpenter2009});} 
\tablefoottext{c}{Chen et~al. (\cite{Chen2005});} 
\tablefoottext{d}{Gautier et~al. (\cite{Gautier2008});} 
\tablefoottext{e}{Kessler-Silacci et~al. (\cite{Kessler2006});} 
\tablefoottext{f}{K\"ohler (\cite{K\"ohler2001});} 
\tablefoottext{g}{Lommen et~al. (\cite{Lommen2007});} 
\tablefoottext{h}{L\'opez-Santiago et~al. (\cite{L\'opez2006});} 
\tablefoottext{i}{Low et~al. (\cite{Low2005});} 
\tablefoottext{k}{Makarov \& Fabricius (\cite{Makarov2001});} 
\tablefoottext{l}{Mamajek et~al. (\cite{Mamajek2002});} 
\tablefoottext{m}{Mamajek et~al. (\cite{Mamajek2004});} 
\tablefoottext{n}{Neuh\"auser et~al. (\cite{Neuh\"auser2003});} 
\tablefoottext{o}{Padgett et~al. (\cite{Padgett2006});} 
\tablefoottext{p}{Rebull et~al. (\cite{Rebull2008});} 
\tablefoottext{r}{Soderblom et~al. (\cite{Soderblom1998});} 
\tablefoottext{s}{Torres et~al. (\cite{Torres2000});} 
\tablefoottext{t}{Wichmann et~al. (\cite{Wichmann2003});} 
\tablefoottext{u}{Zuckerman \& Song (\cite{Zuckerman2004});} 
\tablefoottext{v}{SIMBAD;} 
\tablefoottext{w}{Weise (2007).}}
\end{table}